Author’s response to reviews

Title: A Retrospective Analysis on Metastatic Rate of the Internal Mammary Lymph Node and Its Clinical Significance in Adjuvant Radiotherapy of Breast Cancer Patients

Authors:

Li Li (abcyili@126.com)
Hongyan Zhang (zhy20060901@163.com)
Linwei Wang (wanglinweiwhu@163.com)
Conghua xie (chxie_65@hotmail.com)
Yunfeng Zhou (yfzhouwhu@163.com)
Yahua Zhong (doctorzyh73@163.com)

Version: 4 Date: 14 Feb 2020

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “A Retrospective Analysis on Metastatic Rate of the Internal Mammary Lymph Node and Its Clinical Significance in Adjuvant Radiotherapy of Breast Cancer Patients” (ID: BCAN-D-19-02456R3). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the reviewer’s comments are as following:

Editor Comments:
1. We note that two authors have the same initials (YZ). To distinguish between them in the Authors’ Contributions section please designate them as YZ1 and YZ2, with YZ1 corresponding to Yunfeng Zhou, chosen because this name is furthest up the author list.
A: In the Authors’ Contributions section, we designate the author Yahua Zhong as YZ1 who is the corresponding author, and designate the author Yunfeng Zhou as YZ2.
2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.
A: We have upload our final version of manuscript as a single, final, clean version.

By the way, we have checked our work again and again, and found there were a few mistakes, and the corrections are as following:
1. In the Reference section, the “Zucali RJEJoC” was replaced of “Zucali R” in the reference 20.
2. Owing to our negligence, in the Table 1, the rate of tumor staging IIB was 24.6%, but not 31.6% and the rate of tumor staging IIIC was 13.2%, but not 6.1%. The correct table was uploaded.

We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.
Thank you and best regards.
Yours sincerely
Corresponding author: Yahua Zhong